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• Silicon detectors application tested (histogram and oscilloscope mode). It works, but we still need some reiteration with BI

experts in order to implement all the functionalities of these devices.

• BI experts (M.Duraffourg) have intervened for noise problem in HRS separator semgrid. They have isolated the electronics

and situation have improved. Test to be continued this week.

• Still some problem with new HRS separators wire scanners. We can’t use 2 at the same time (noise cross talk issue) and

one of them seems not working properly. To be addressed this week.

• Mass scan application tested. Still some small adjustment to be performed.

• First setting up of HRS RFQ. Gas regulation system seems to be working properly. We have stopped because a foreseen

intervention by EN-EL (refurbishment of ISOLTRAP electrical network) have created a general power cut for all the ISOLDE

hall sectors vacuum controls. Still 4 sectors not restarted. Vacuum people very active (thanks!).

• GPS setting up done at 30kV for Collaps in order to commission their experimental station.

• This week, by today evening, we stop both target heating in preparation of intervention on the ventilation and cooling

water system (3 days).

• HIE Isolde cool down continuing according to plans (cavities at 170K).


